

The relationship between internalised weight bias and biopsychosocial outcomes in children and youth: a systematic review



Tiarna Foster¹, Melissa Eaton¹ and Yasmine Probst^{1*}

Abstract

Objective To synthesise the evidence on the relationships between internalised weight bias (IWB) and biopsychosocial health outcomes in individuals ≤ 25 years.

Methods A systematic review was conducted by searching five scientific databases up to May 2022 to retrieve studies that investigated associations between IWB and biopsychosocial outcomes. Articles with participants ≤ 25 years, at least one validated measure of IWB, one measure of a biopsychosocial outcome, and were observational were included. Excluded articles involved systematic literature reviews, case study reports, intervention studies, meta-analyses, grey literature, pilot, and feasibility studies. Quality assessment was carried out using the American Dietetic Association Quality Criteria Checklist. The protocol was registered with PROSPERO, ID number CRD42022323876.

Results Two hundred and sixty-six articles were identified. Nineteen were eligible for inclusion, (15 cross-sectional and 4 prospective). The Weight Bias Internalization Scale and the Weight Self-Stigma Questionnaire were the most used tools to assess IWB with large heterogeneity in tool types used to assess biopsychosocial measures. IWB had positive associations with psychopathology, eating disorder symptomology, higher BMI, being female, and experiences of weight stigma. It was negatively associated with quality of life, body image, physical activity, social ability, self-esteem, and socioeconomic status.

Discussion IWB associated with adverse biopsychosocial outcomes in children and youth populations. IWB may be more clinically relevant in assessing at-risk children and youth than physical weight due to its psychosocial aspects and ability to expand beyond the scope of BMI. Research would benefit from better assessment tools designed for children and youth that accurately measure IWB. Future research should focus on increased diversity and longitudinal study designs with children and youth-specific populations.

Keywords Internalised weight bias, Weight bias, Weight stigma, Internalised weight stigma, Youth, Eating disorders, Weight bias internalisation

Plain English Summary

The objective of this systematic review was to bring together the current evidence on the relationship between internalised weight bias (IWB) and health outcomes in individuals under the age of 25. The systematic review was conducted by searching five scientific databases to retrieve studies that investigated associations between IWB

*Correspondence: Yasmine Probst yasmine@uow.edu.au Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

and biopsychosocial outcomes. Two hundred and sixty-six articles were identified, with nineteen eligible for inclusion into the review. IWB was associated with increased mental illness, eating disorder symptoms, BMI, being female, and experiences of weight stigma. It was associated with a decreased quality of life, body image, physical activity, social ability, self-esteem, and socioeconomic status. Overall IWB was found to be associated with negative health and social outcomes in children and youth populations. Future research in this area should focus on increased diver-

sity, longitudinal study designs and designing children and youth specific tools that accurately measure IWB.

Public significance statement

Results from this study revealed that significant associations exist between internalised weight bias and negative biopsychosocial outcomes in a population below 25 years of age. The results add to the growing body of literature, with findings helping to inform health care professionals of the importance of screening for internalised weight bias and recognising its signs in children and youth. It may also provide guidance when developing individual and population level nutrition interventions and educational programs.

Introduction

Weight bias is a form of prejudice that leads to stigmatisation, social rejection and the devaluation of individuals who do not comply with social norms for weight [1, 2]. This can transform into internalised weight bias (IWB) when individuals become aware of, and internalise these conceived notions, proceeding to mistreat or devalue themselves due to a perceived self-classification of overweight (OW) [3]. Research related to IWB is gaining attention due to evidence showing an association with negative health outcomes socially, psychologically, and physically. A study by Puhl et al. reported that 44% of Americans experience IWB [4] with presently no prevalence data available for Australians.

Studies have demonstrated consistent links between IWB and adverse psychological and physical health outcomes within adults [5–7]. IWB has been identified, through cross-sectional studies, to increase adverse mental health indices such as depression, anxiety, poor body image, health-related quality of life, and eating disorder symptomology, even when controlling for BMI, with it playing a key role in psychosocial health [8, 9]. Though there are fewer studies examining physical health, associations have been observed between IWB and a higher BMI, lower self-efficacy for engaging in health-promoting behaviours, and poor dietary adherence in treatment-seeking populations [10].

IWB does not exist separate to external weight stigma, but rather is a product of it and the immediate and wider social environment [4]. Lived experiences of external weight stigma occur through our social networks and contribute to the multifactorial nature of IWB. Experiences from the home environment, and peers, such as weight-based teasing have all been reported to contribute to the development of IWB [11], with children and youth reporting higher levels of weight based teasing and vulnerability [12].

The influence of a child's environment in addition to their life experiences are vital to consider when addressing any form of stigma. The prevalence of eating disorders is highest between the ages of 12 and 25 years, though weight bias and dieting behaviours may appear as young as 5 years of age [13, 14]. Children are more dependent on their environment for social feedback making them vulnerable to the negative consequences of IWB [15, 16]. This period of development is one in which a self-depreciating mindset and perception of oneself are is presumed to appear with long-term impacts and beliefs likely to track into adulthood [17, 18]. Angel. et al., demonstrated that children and youth who utilised extreme weight control behaviours in early adolescence demonstrated an 8.4-20.4% increase in these behaviours as they entered young adulthood [18], highlighting the need for early intervention. Through greater research and early intervention, health professionals can aid in preventing the chronic nature of these developed beliefs and address them before they compound into adulthood [19].

Research to date within populations under and over 25 largely investigates IWB in treatment seeking populations, with research showing that these populations are at a higher risk for scoring lower on health related quality of life tests when compared to peers within the community in regard to IWB [20, 21]. This has then impacted on their treatment/health outcomes. A study assessing IWB scores with a "lean" population (mean: 22.28 ± 1.89 kg/m², range 15.80-24.98 kg/m², using the classification of 'normal weight' or 'underweight' $(BMI < 25 \text{ kg/m}^2)$ according to the National Institutes of Health's weight classification guideline) found that 38.6% of respondents self-reported as 'overweight' or 'obese' [22]. They proceeded to find that respondents who believed themselves to be 'OW/obese' had significantly greater IWB scores than those who accurately perceived themselves as 'normal weight'. In addition,

formerly 'overweight' persons may continue to internalise weight bias even after weight loss [23], hence the importance of continued monitoring for IWB across the weight spectrum.

Further, to better understand the relationship between health, experiences of weight stigma and weight, it is important that we understand its mediators. IWB demonstrates a mediational relationship [24, 25] between weight bias and adverse health outcomes (Fig. 1), with evidence highlighting a potentially damaging impact on individuals regardless of BMI and experiences of stigma. This highlights a need to look beyond physical weight when addressing IWB and address biopsychosocial outcomes as they demonstrate a reciprocal predicative relationship with IWB; whereby increasing IWB increases adverse outcomes and vice versa [26]. For example, an increase in IWB may lead to decreased self-efficacy and motivation to engage in health-promoting and social behaviours, which may in turn negatively influence social and physical outcomes [27].

Health outcomes are vital to explore within child and youth with early experiences creating biological 'memories', that shape development, and set the foundations for lifelong health [28]. Psychosocial problems in children can lead to serious consequences for social competence, learning, and lifelong physical health outcomes. Distinguishing between children and adolescents is also of importance with clear cognitive, physical, and psychosocial differences occurring between the age groups due to increased vulnerability related to hormonal changes, changes in appearance, and transition toward adult roles and responsibilities [29]. Changes associated with puberty are a required consideration when assessing children and youth's self-image and worth due to increased vulnerability in this population. This population is vital to investigate independent to adult populations as children and youth differ cognitively, physically, and psychosocially, with results unable to be generalised. Many lifelong adverse health behaviours are believed to originate in the earlier years before compounding into



Fig. 1 Mediational relationship between weight bias, internalised weight bias, and adverse health outcomes

adulthood, hence a need for greater research to allow for early intervention [17–19].

This systematic review aimed to synthesise and evaluate the evidence on the relationship between IWB and biopsychosocial health outcomes of individuals' \leq 25 years of age to identify associations and limitations within the literature. It aims to highlight key research gaps that can inform future studies around IWB. The findings may help to inform health care professionals of the importance of screening for IWB and recognising its signs in child and youth populations.

Methods

Search strategy

A systematic literature search was conducted to identify studies that investigated an association between IWB and any biopsychosocial outcomes (e.g. BMI, depression, eating disorder symptomology) in individuals' \leq 25 years old. This systematic review followed the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) and the Cochrane handbook for systematic review guidelines and was reported according to the PRISMA 2020 checklist [30] (Fig. 1). The protocol for this search is registered with PROSPERO, ID number CRD42022323876.

The literature search was conducted April-May 2022 using five health-related scientific databases: PsycInfo, Web of Science, Cinahl, Medline (EBSCOhost) and Scopus. Database and keyword selection was informed by previous studies of IWB and through pilot searches. A preliminary search of the literature was used to inform the choice of key words and MeSH terms. The final search comprised the following keyword combinations: child, *adolescen, * teenager, * youth, "young person," "under 18," "under eighteen," "school aged," "university student,* "preschool children," "under 25," "under twentyfive," intervention, * "trial," rct, observation, * cohort, longitudinal, cross-sectional, prospective, retrospective, quasi-experimental, "weight bias" "weight-bias," "weight discrimination," "weight based discrimination," "weightbased discrimination," "weight prejudice," "weight stigma," "weight-stigma," "weight bias internalization," "weight bias internalisation." Keywords and boolean combinations were applied to each of the databases and adapted to suit database specific needs only.

Eligibility criteria

Articles published in peer-reviewed journals with participants \leq 25 years that included at least one validated measure of internalised or self-directed weight bias (e.g. Weight Bias Internalization Scale or Weight Self-Stigma Questionnaire), in addition to at least one biopsychosocial outcome variable (e.g. diagnosed or self-reported anxiety or depression, self-esteem, disordered eating, body weight, BMI, physical activity etc.) were included. Only observational quantitative study designs (e.g., crosssectional, longitudinal) were included and no limitations were placed on the year of publication. The term youth was used to include both adolescents and youth, and encompasses the ages 13–25 years. Children was used to refer to participants \leq 12 years. In incidences when both youth and children are referred to, the phrase a population under the age of 25 was utilised. There was no predetermined lower limit for inclusion of age.

Exclusion of studies was informed for the Cochrane handbook and previous studies [10]. Excluded studies included systematic review and meta-analyses studies, case study reports, intervention studies and meta-analyses, grey literature, pilot and feasibility studies. Reference lists of these articles were manually searched and cross-checked against eligibility criteria to ensure eligible papers had not been missed.

Article selection

The search outcomes were imported to Covidence for duplicate removal prior to title and abstract screening against the eligibility criteria by one reviewer (TF). Full text screening was completed by three reviewers (TF, YP, ME) to exclude studies that did not include a measure of internalised or self-directed weight bias and a biopsychosocial measure. The proportionate agreement was 63% and 59% between reviewers, with all studies reviewed independently by two researchers before being included in the review.

Data abstraction and synthesis

The following data was extracted from all included reports: country where data was collected, publication year, author/s, study design, population information (e.g., size, source, etc.), sample demographics (e.g., age, gender, BMI, % BMI category if available, income, parental education level, and ethnicity when available etc.), measures used to assess internalised weight stigma and measures used to assess a biopsychosocial outcomes and outcome data. Studies were divided into three categories: (1) psychological outcomes; (2) physical outcomes; and 3) social outcomes.

Where articles reported multiple biopsychosocial outcomes, articles were not categorised as one single outcome but rather each outcome was individually categorised to each category. Therefore, the total number of outcomes may exceed the total number of included studies and studies may appear across multiple categories. Finally, for instances where multiple outcomes followed the same theme they were categorised under the same definition. For example, all eating disorder symptomology outcomes were categorised under one category due to the heterogeneity of the data.

Quality assessment

Publication bias was assessed by one researcher (TF) using the American Dietetic Association Evidence Analysis Library Quality Criteria Checklist [31]. As per Cochrane guidelines, quality assessment was conducted following inclusion of studies to allow for inclusion to be guided by the inclusion criteria and avoid influence. This provided insight into various biases related to each of the studies.

Due to the limited research designs (e.g., longitudinal or experimental), criteria were modified based on prior research on IWB. Quality assessment of studies included in this review addressed: cross-sectional versus longitudinal study design; validity of biopsychosocial outcomes measure: self-report versus objective measurement; and validity of the IWB measure.

Results

Figure 2 depicts the selection flow as per PRISMA guidelines. Across the databases, after duplicates were removed, 266 articles were identified. Six additional articles were identified during manual screening of reference lists. Following quality assessment, 19 studies remained for data extraction.

Characteristics of included studies and participants

Of the 19 studies included, 21% (n=4) were prospective studies and 79% (n=15) were cross-sectional. Fifteen relied on retrospective data analyses. Ten studies were included from Eurocentric countries (Germany [21, 32-34], Australia [24], USA [35-38], Canada [32]), six from Asian (China [26, 39], Hong-Kong [26], Korea [40]) and three from Middle Eastern countries (Iran [41, 42], UAE [43]). A majority of studies (68.4%) were published between 2019 and 2021. The earliest study included was published in 2012.

Only six studies provided information on the ethnicity of participants. A majority (69.83%) of identified ethnicities were white. Only, 14.53% were identified as black, 10.76% as Hispanic or Latino, 8.29% as Asian, and 5.80% were identified and classified as other by the study authors. Of the participants, 61.45% were female, and 38.55% were male, with four studies exclusively investigating a female population. Three studies [21, 32, 33] investigated the role that sex played on IWB scores and found females to have higher IWB scores, whilst two found no difference [35, 38]. Lastly, the ages of participants ranged from 5 to 25 years. Three studies involved dyads of one parent and child [39].



Fig. 2 PRISMA flowchart of article screening and selection process

Tools used to measure internalised weight bias

To assess IWB, studies predominantly used the Weight Bias Internalization Scale (WBIS; n=16, 84.21%) or a modified version and the Weight Self-Stigma Questionnaire (WSSQ; n=5, 26.32%). One study that utilised the Weight- and Body-Related Shame and Guilt Scale (WEB-SQ) was included in this review due to it explicitly stating that it was used as a measure of IWB. Seven of the 19 studies modified the Weight Bias Internalization Scale: four used a modified WBIS (WBIS-M), two a child WBIS (WBIS-C), one a WBIS tool for youth (WBIS-Y) [26, 32, 44], Table 1.

Table 1 summarises all included studies on IWB and outcomes in children and youth. All studies included at least one validated measure of a biopsychosocial outcome, with high variability in the types of tools used.

Psychological outcomes

All nineteen studies found positive associations between higher levels of IWB and at least one negative psychological outcome [6–28, 31, 34–36, 38–40]. The findings are summarised in Fig. 3 which demonstrates the associations between IWB and psychological distress, eating disorder symptomology, self-esteem, and quality of life as IWB scores increased.

Positive associations were seen between prevalence of eating disorder symptomology (binge eating, eating to cope, eating restriction, bulimia, oral control, restrained eating, uncontrolled eating, emotional eating), and pathopsychological (emotional problems, depression, anxiety, psychological distress) as IWB scores increased.

IWB was significantly and positively associated with elevated levels of psychopathology and negative

			5		<u>`</u>					
Reference	Study design	n (% females)	Mean age, years	Ethnicity	Reported BMI, kg/m ²	Measure of	IWS mean	Relationship		
	(country)		(ralige)			GWI		Psychological	Physical	Social
Ahorsu et al. [41]	Prospective (Iran)	1497 (54.30%)	15.10 ±6.00		baseline = 31.8 ± 5.4 six-months = 33.8 ± 5.1	VBIS	3.8 (±0.97)	***IV/S versus binge eating at follow-up: r=0.22, $p=<0.01IWS versus psy-chological distressat follow-up:r=0.15$, $p=<0.01MWS versus foodaddictionp=<0.01$, $p=<0.01$,		
Guptitu-Plath et al. [21]	Cross-sectional (Germany)	191 (51.31%)	(13–19)	German: 49.74% Other: 49.74%	BMI 26–29 kg/ m² = 15,71% BMI> 30 kg/ m² = 33.51% BMI> 35 kg/ m² = 47,74%	WBIS-Y	Girls: 4.01 Boys: 3.54	*Self-esteem: a = -0.70 Internal related LoC: $a = -0.01$ External related LoC: $a = 0.28$ Self-efficacy: a = -0.39 generic health- related quality of life: $a = -0.50$	*BMI-SDS: a = 0.20 obesity-specific QoL: a = - 0.77	
Fan et al. [39]	Cross-sectional (China)	430 dyads (44%)	1007 ±1.42; (8−12)		18.47 ± 4.16	WBIS; WSSQ	WBI: 1.59 (0.56) WBI Boys: 1.50 (0.53) BMI> 25 kg/m ² : 1.47 (0.48) BMI ≤ 25 kg/m ² : 2.06 (0.62)	**Perceived weight stigma: 0.81, p < 0.001	***5ex: r= 0.06 BMI: r=0.025, p< 0.001	***Weight stigma: r= 0.39, p< 0.001
Gmeiner et al. [32]	Prospective (Germany)	1463 (51.74%)	8.35,±0.94; (6–11)		BMI< 18.5 kg/ m ² = 6.08% BMI 18.5 - 25 kg/ BMI > 25 kg/ BMI > 25 kg/ m ² = 3.19%	WBIS-C	Girls: 1.68 (0.62) Boys:1.56 (0.56) BMI> 25 kg/m ² : 2.16 (0.68) BMI ≤ 25 kg/m ² : 1.53 (0.52)	***Body dissatis- faction: $r = 0.36$, p < 0.001 Relevance of one's own figure: $r = 0.2$, p < 10.001 : r = -0.05, $p = 0.046Depressivesymptoms r = 0.24,p < 0.001$	***BMI-5D5: r= 043, p< 0.001	***Weight-related teasing: r = 0.28, p< 0.001
Gmeiner et al. [34]	(Germany)	1061 (52.1%)	11±0.9; (9−13)		BMI < 18.5 kg/m² t: 8% BMI 18.5 – 25 kg/m²: 78.3% BMI 25 – 30 kg/ m²: 8.1% BMI > 30 kg/m²: 5.6%	WBIS-C	1.55±0.55) BMI> 25 kg/m². 2.09,±0.66 BMI≤ 25 kg/m². 1.46±0.47	111 WBIS-C versus SDQ: AUC = 0.67, p < 0.001 WBIS-C versus SCOFF: AUC = 0.77, p < 0.001		

Table 1 Summary of included studies with outcomes categorised as psychological, physical and social factors

Reference	Study design	n (% females)	Mean age, years	Ethnicity	Reported BMI, kg/m ²	Measure of	IWS mean	Relationship		
	(country)		(range)			IWB		Psychological	Physical	Social
Himmelstein et al. [38]	Cross-sectional (USA)	148 (50%)	15.97±1.25	White: 90.6% Black: 2.0% Asian: 2.7% Latino: 4.7%	27.06±4.39	W-SI8W	5.45 (0.88)	***WBI versus respond- ing to bullying with negative emotions; $\beta = 0.62$, $p = < 0.001$ WBI versus respond- ing to bullying with indifference: $\beta = -0.07$, $p = 0.487$ WBI versus coping with solutions with endifference: $\beta = -0.07$, $p = 0.07$, $p = 0.001$ with eating; $\beta = 0.70$, $p = < 0.001$		
Kamolthip et al. [26]	Cross-sectional (Hong-Kong)	437 dyads (43.90%)	10.07 ± 1.43; (8–12)		18.59±4.25	WBIS; WSSQ	Ative group: r = 22.51 (8.04) Non-active group: r = 24.10 (8.62) WSSQ active group: r = 22.67 (8.58) WSSQ non-active group: r = 24.34 (9.36) Self-stigma active group: r = 34.54 (11.67) self-stigma non-active group: r = 37,13 (12.67)	**Physical activity versus self-stigma r = -0.11, p = .029 Self-stigma versus Kid-KINDL versus Kid-KINDL rid-ated: r = -0.38, p < .001 Self-stigma ver- sus SNU $r = -0.60,$ p < .001 self-stigma ver- sus SNU $r = -0.34,$ p < .001		
Lin et al. [42]	Cross-sectional (Iran)	934 (52.46%)	15.7±1.2; (13-18)		33.0±4.7	WBIS	3.0±0.5	¶ DASS-21: r=0.38 (<i>p</i> <0.001)	¶ zBMI: r=0.23 (p<0.001)	

erence Study	y design	n (% females)	Mean age, years Ethnicity	Reported BMI, kg/m ²	Measure of	IWS mean	Relationship		
	1		(range)				Psychological	Physical	Social
ano et al. [45] Cross (Cana	s-sectional add)	156 (48%)	16.31 ±0.85 (14-19)	BMI 26–29 kg/m² (76.3%) BMI ≥ 30 kg/ m² (23.7%)	OS MM	Self-devaluation: 0.778 fear of enacted stigma: 0.828	**WSSQ (self- devaluation) versus Self-esteem: r = -0.437 , $p < 0.001$ WSSQ (self-devalu- ation) versus Physi- cal appearance: r = -0.428 , $p < 0.001$ WSSQ (self-devalu- ation) versus Anxi- ety: r = 0.242 , p < 0.01 WSSQ (self-devalu- ation) versus Depression: r = 0.242 , p < 0.01 WSSQ (self-devalu- devaluation) versus Eat- ing-related control: r = 0.342 , $p < 0.001$ WSSQ (self-devalu- ation) versus Eat- ing-related control: r = 0.342 , $p < 0.001$ WSSQ (self-devalu- ation) versus Eat- ing-related guilty: r = 0.429 , $p < 0.001$ WSSQ (self-devalu- ation) versus Eat- ing-related guilty: r = 0.429 , $p < 0.001$		
								ation) versus Eat- ing-related guilty: r =0,429, p < 0.001	ation) versus Eat- ing-related guilty: r = 0.429, $p < 0.001$

Reference	Study design	n (% females)	Mean age, years	Ethnicity	Reported BMI, kg/m ²	Measure of	IWS mean	Relationship			
	(country)		(ralige)					Psychological	Physical	Social	
O'Brien et al. [24]	Cross-sectional (Australia)	634 (73.40%)	19.70±3.07	White: 60% Asian or pacific islander: 37% Black: 3%	2240±4.14	WBIS-M	BMI> 25 kg/m ² : 3.91 (1.65) BMI≤ 25 kg/m ² : 2.77 (1.29)	**WBIS versus BMI: r=0.15, $p < 0.001$ WBIS versus Gen- der: r=0.23, p < 0.0001 WBIS versus Stigma total: r=0.56, p < 0.0001 WBIS versus DASS- 21: r=0.51, p < 0.0001 WBIS versus Emo- tional eating: r=0.52, $p < 0.0001$ WBIS versus Emo- tional eating: r=0.52, $p < 0.0001$ WBIS versus LOCES- sus Uncontrolled eating: r=0.65, p < 0.0001 WBIS versus LOCES- B: r=0.65, p < 0.0001			
O'Hara et al. [43]	Cross-sectional (UAE)	420 (100%)	23.12 ±4.62		24.09 ± 5.77	WEB-5G	20.15 (11.65)	**WEB-SG ver- sus BMI: r= 0.47** WEB-SG versus EAT- 26: r= 0.43** WEB-SG versus AF: r= 0.30** WEB-SG versus AFA: r= 0.39** WEB-SG versus AFA: r= 0.43** WEB-SG ver- sus SAAS: r= 0.56** WEB-SG versus SATAQ: r= 0.35** ** A<001			

lable i (cont	lineu/										
Reference	Study design	n (% females)	Mean age, years	Ethnicity	Reported BMI, kg/m ²	Measure of	IWS mean	Relationship			
	(country)		(range)			GWI		Psychological	Physical	Social	
Pakpour et al. [46]	Cross-sectional (China)	287 (46.70%)	10.21 ± 1.31 (8–12)		BMI≤ 25 kg/m²: 66.2% BMI> 25 kg/m²: 33.8%	WSSQ; WBIS	191 ±0.75 WBIS: 2.28 ±0.64	**WBIS versus PWD: r=0.35, $p < 0.001$ WBIS versus Kid- KINDL: r = -0.32, p < 0.001 WBIS versus Sizing WBIS versus Sizing WBIS versus Sizing p < 0.001 WSSQ ver- sus PWD: r = -0.31, p < 0.001 WSSQ versus Kid-KINDL: r = -0.37, $p < 0.001$ WSSQ versus Sizing me up: r = -0.39, p < 0.001			
Pik Chu Wong et al. [44]	Cross-sectional (Hong-Kong)	124 dyads (40%) OW: 9.36 ± 1.17; 8-12 Non- OW = 9.73 ± 1.28	(8–12)		BMI> 25 kg/ m² = 22.86 ± 2.32 BMI ≤ 25 kg/ m² = 16.27 ± 2.10	WBIS; WSSQ	BMI = 25 kg/ m ² group: 26.49±8.68 BMI > 25 kg/m ² group 21.58±754 WSSQ BMI = 25 kg/m ² 14.50±4.89 WSSQ BMI > 25 kg/m ² 11.02±4.37	**CW group WBIS versus BSRS-5: r = 0.525, p < 0.001 WSSQ versus BSRS- 5: r = 0.336, p < 0.001 Non-OW group WBIS versus BSRS-5: r = 0.376, p < 0.001 WSSQ versus BSRS-5: r = 0.298, p < 0.001			

Reference	Study design	n (% females)	Mean age, years	Ethnicity	Reported BMI, kg/m ²	Measure of	IWS mean	Relationship			
	(country)		(range)			1WB		Psychological	Physical	Social	
Puhl et al. [35]	Cross-sectional (USA)	15.26±1.63	(13–18)	White: 90.5% Black: 2% Asian: 2.7% Latino: 4.7%	27.06±4.39	VBIS	5,45±0.88 Girls: 5,40±0.97 Boys: 5,49±0.78 BMI ≥ 25 kg/m ² : 5,40*±0.59 BMI > 30 kg/m ² : 5,85*±0.92 5,85*±0.92	***WBI verses the subble verses beinge eating: $\beta = -0.18$ WBI versus Eating to cope: $\beta = 0.50$ WBI versus Female: $\beta = 0.04$ WBI versus BMI; $\beta = 0.04$ WBI versus Race: $\beta = 0.02$ WBI versus Race: $\beta = 0.02$ WBI versus Race: $\beta = 0.24$ WBI versus Fre- quency of mother's comments about their own weight $\beta = 0.20$			
Ra et al. [40]	(Korea) (Korea)	233 (100%)	13.43 ± 0.68; (12–14)		53.92±30.30	WBIS	0.08±1.01	***Fear of obe- sity: β = .43, <i>p</i> < .001		Attachment to teacher: $\beta =11, p = .029$ Perceived socio- cultural pressure $\beta = .34, p < .001$	

Foster et al. Journal of Eating Disorders (2024) 12:38

Table 1 (cont	inued)										
Reference	Study design	n (% females)	Mean age, years	Ethnicity	Reported BMI, kg/m ²	Measure of	IWS mean	Relationship			
	(country)		(range)			SWI		Psychological	Physical	Social	
Roberto et al. [36]	(USA) (USA)	57 (80.70%)	1565 ± 1.08; (14-18)	White: 50.9% Hispanic: 31.6% American: 12.3% Other: 5.3%	(35.90-72.30) (35.90-72.30)	WBIS	4.29±1.52	**WBIS versus age: r=0.169 WBIS versus Beck Depression Inven- tooy: r=0.520, p<0.01 WBIS versus Youth Self-Report Total: r=0.310, p<0.05 WBIS versus Paedi- aric Quality of Life Tene Self-report: r=0.483, p<0.01 WBIS versus Mul- tidimensional Anxiety Scale for Children: r=0.467, p<0.01 WBIS versus Mul- tidimensional Anxiety Scale for Children: r=0.467, p<0.01 WBIS versus Shape concern Subscale: r=0.570, p<0.01 WBIS versus Shape concern Subscale: r=0.570, p<0.01 WBIS versus BMI: r=0.044			

Romano et al. [37] Cross-sectional Sc (USA) (15A) 22 Sc Sc Sc Sc Sc Sc Sc Sc Sc Sc Sc Sc Sc		(range)			IWR				
Romano et al. [37] Cross-sectional 5a (USA) 22 5a 57							Psychological	Physical	Social
	sample 1; 1228 75,8%) 22,27 ± 5,83 sample 2 = 1368 75,80%)	20.60 ± 3.47	White: 41.2% Black or African American: 37.8% Multiracial: 12.7% Asian Asian American, Asian Native Hawaiian, or Pacific Islander: 5.0% Other race: 2.7% Other race: 2.7% Hispanic or Latinx: 14.4%	26.47±6.57	M-SIBW	BMI < 25 kg/m ² : 2:59 (1.27) BMI 26-29 kg/m ² group mean: 3.39 BM1-30 kg/m ² group mean: 4.07 (1.57)	If WBI versus BMI r = 0.385, p < 0.001 WBI versus Expe- rienced weight stigmar = 0.440, WBI versus Body dissatisfaction: r = 0.740, p < 0.001 WBI versus Binge eating: r = 0.439, p < 0.001 WBI versus Purging: p < 0.001 WBI versus Exces- sive exercise: 0.095, p < 0.01 WBI versus Exces- sive exercise: 0.095, p < 0.01 WBI versus Muscle bing versus Muscle wBI versus Musc		
Saffari et al. [47] Cross-sectional 36 (China)	91 (100%)	22.85 ± 4.03		21.30±3.53	wsso	30.49 (10.58) High physical activity group: 30.40 (10.01) Moderate physi- cal activity group: 26.37 (10.82) Low physical activity group: 31.74 (10.33)	MRSS ver- sus Age: r = 0.04 (0.40) WRSS ver- sus Weight status: r = 0.40 (< 0.001) WRSS versus Psy- chological distress: r = 0.45 (< 0.001)		
Zub et al. [33] Prospective 10 (Germany)	1047 (52%)	Time 1:9.0±0.9; (7–11) Time 2: 10.5±0.97; (7–11)		BMI<18.5 kg/ m²=6.3% BMI 18.5-25 kg/ m²=81.4% BMI>25 kg/ m²=12.4%	WBIS-C	WBI5:2333 (8.33) WSSQ Q1-6 score: 12.51 (4.70) WSSQ Q7-12 score: 10:94 (4.88)	**Perceived Weight stigma: r=0.81 Child related KID- KINDL: r=-0.38	**BMI: r = 0.25	**Experienced weight stigma: r = 0.39 Sizing me up: r = - 0.60 Sizing them up: r = - 0.34

UP (child-rated report) and *STU* Sizing Them Up (parent-rated report), *SABAS* Smartphone Application-Based Addiction Scale, SAAS Social Appearance Anxiety Scale, SATAQ-ED Sociocultural Attitudes Toward Appearance Questionnaire for people with Eating Disorders, *SSI-B* Stigmatizing Situations Inventory—brief, *SDQ* Strengths and Difficulties Questionnaire, *TFEQ-R18* Three Factor Eating Questionnaire, *YFAS-CKid-KINDL* Yale Food Addiction Scale for Children, *YSR* Youth Self-Report, *WSS* Weight Self Stigma Eating Disorder Examination Questionnaire, *FPSI* Eating Pathology Symptoms Inventory, *EDD5* Everyday Discrimination Scale, *EWS* Experienced Weight Stigma, *FNAES* Fear of Negative Appearance Evaluation Scale, *DTGA* German Depression Test for Children, *KAT* German Anxiety Test for Children, *HADS* Hospital Anxiety and Depression Scale, *LOCES* Loss of Control of Eating Scale, *MASC* Multidimensional Anxiety Scale for Children, *PEDS-QL* Paediatric Quality of Life Inventory 40, *POTS* Perception of Teasing Scale, *PANAS* Positive and Negative Affect Schedule, *RSE* Rosenberg Self-Esteem Scale, *SDQ-II* Self-Description Questionnaire II, *SMU* Sizing Me

*Crohnbach, **Pearsons correlation, ***Regression, ¶Mediation, ¶¶Bivariate correlation, ¶¶¶ROC analysis

Psychological										
Psychological distres	5							Quality o	of Life	
Psychopathology	Anxiety	Depression	External	Eating disorder	Body	Self-	Self-	Health/	Physical	Social
			LoC	symptomology	image	efficacy	esteem	generic		
\uparrow	\uparrow	\uparrow	\uparrow	\uparrow	\downarrow	\downarrow	\downarrow	\downarrow	\downarrow	\downarrow
(21, 24, 33-35, 38, 40-42, 44,	(24, 36, 45, 47)	(24, 32, 36, 45) (47)	(21)	(24, 33-36, 38, 41, 43, 45)	(21, 35, 36, 40,	(21)	(21, 44, 45)	(21, 26, 36,	(36, 42, 44)	(39, 42, 44)
45, 47, 48)					41, 47)		\uparrow	39, 42-44,		
							(32)	47)		
Physical										
Physical activity	Sex (girl)	Higher BMI								
\downarrow	\uparrow	\uparrow								
(26, 43, 48)	(21, 32, 33)	(21, 24, 32-36, 38, 47)								
	No difference									
	(35, 38)									
Social										
Experience weight	Sociability /	Socioeconomic								
stigma	behaviours	status								
	issues									
\uparrow	\downarrow	\downarrow								
(24, 32, 33, 37-39, 41, 44)	(36, 38, 40, 44)	(32)								
			_							
1 study	3	4		5 6		8		9	13 studi	ies
Legend: Col	our scale indicat	ing the number of	studios (cito	d by reference numb	er) reporting	the same di	rection of rel	ationshin		

Fig. 3 Heat map indicating the relationships reported in included studies of IWB. Arrows indicate the direction of relationship with IWB as scores increase. Abbreviations: *BMI* Body Mass Index, *IWB* Internalised weight bias, *LoC* Locus of control. Legend: Colour scale indicates the number of studies (cited by reference number) reporting the same direction of relationship

psychosocial outcomes [33, 36, 41, 45] across sex and weight groups [33, 41]. IWB mediated the relationship between elevated BMI and quality of life [39], and psychosocial problems [21, 33, 48], with IWB shown to be more important than weight status in explaining psychological functioning [33] via cross-sectional studies. Five studies found IWB significantly and negatively correlated with health-related quality of life [21, 26, 43, 44].

Physical outcomes

Fourteen studies investigated associations between BMI and IWB. There was heterogeneity in the data for how BMI was measured. Six articles supplied a mean and standard deviation, whilst six recorded the percentage of weight categories (e.g., 'underweight', 'overweight'), and seven reported both. Fifteen (78.95%) studies relied on self-reported data for weight. Six articles only investigated OW or children and youth with a BMI \geq 30 kg/ m² [21, 36, 41, 42, 45, 46]. Two of the nine studies that explored the weight spectrum had 'normal'/'underweight' participants as a minority, with four of these studies combining the 'underweight' and 'normal' weight groups [35, 44, 46, 47]. Nine articles found a positive association between IWB and BMI [21, 24, 32-36, 38, 46]. 'Overweight' compared to 'non-overweight' groups showed higher levels of IWB [24, 36, 39, 44], eating disorder symptomology [24], and decreased quality of life [36, 44], with IWB demonstrating a mediational relationship between physical weight and quality of life [39], Fig. 4.

Two studies [26, 47] investigated the association between IWB and physical activity engagement, finding a statistically significant negative correlation, with IWB scores being lower in active groups when compared to non-active groups.

Social outcomes

Ten studies investigated social outcomes. Five studies [36, 38, 40, 44] found a positive association between IWB and negative social characteristics in individuals (e.g. conduct issues, attachment to teachers, avoidance behaviours, maladaptive behaviours). One study [32] found that higher IWB scores and lower SES were positively associated. Lastly, eight studies found experiences of experienced weight stigma to have a strong positive association with IWB [24, 32, 33, 37–39, 41, 44].

The findings of this literature are summarised in Table 1 with the demonstrated correlations between IWB. Figure 3 depicts the association between biopsychosocial outcomes and IWB as IWB scores increased. Further mediational relationships can be found in Fig. 4.

Quality assessment

The quality assessment found three studies to be neutral, and 16 to be positive, indicating good quality of data. However, there was a lack of robust data due to the observational nature of included articles.

Fig. 4 The suggested mediational relationships between IWB and biopsychosocial outcomes. A Increased BMI and decreased physical [42], weight-related [49], health-related [39], and psychosocial QoL[42], increased BMI and increased psychopathology [42] and emotional problems [33] and increased BMI and decrease physical activity levels [47]. B Increased weight teasing and increased eating symptomology [24, 33] and emotional problems [33]. C Decreased physical activity and decreased health related QoL

Discussion

This systematic review synthesises the evidence on associations between IWB and biopsychosocial health outcomes in individuals ≤ 25 years. To our knowledge, this review is the first conducted about IWB in individuals ≤ 25 years and confirms that this topic has received limited attention, with a lack of consistency in methods across studies. Despite this lack of coherence, clear associations have been demonstrated between IWB and higher severity of psychological distress, eating disorder symptomology, reduced motivation/self-efficacy to engage in health-promoting behaviours, (e.g., physical activity), higher BMI and decreased positive sociability. Additionally, experiences of weight stigma demonstrated one of the strongest positive associations with IWB, with IWB scores increasing as experiences of weight stigma increased.

Tools for measuring IWB

This review highlights the need to adopt more objective measures of IWB to enhance confidence in study findings and reduce bias. Eighteen of the 19 studies utilised two self-reporting tools (WBIS and WSSQ) to assess IWB. Although both tools have been shown to demonstrate good psychometric properties and validity, have been translated into several languages, and demonstrate good internal consistency [21], their reliance on self-reported data leaves room for misinterpretation and reporting bias. Furthermore, studies that utilised the WBIS and WSSQ were inconsistent in how they modified tools creating variability in how the scale was used and interpreted. A systematic review of IWB [10] with adults presents the question of how well these tools measure internalisation of weight stigma. For example, the WSSQ measures anticipated stigma from others, which has been differentiated from internalised stigma [10]. Other studies suggested the link between IWB and psychological factors could be attributed to some questions overlapping with other self-judgement factors such as self-esteem and body image [50].

Further, caution should be taken when studying sensitive topics with children, a vulnerable population, to ensure distress is not created for participants. Ciupitu-Plath., et al., found that 4 out of 10 interviewees' feedback highlighted potential stress that may be caused to participants when filling out the modified WBIS tool. Paediatric healthcare providers also believe that the engagement in weight talk with children and youth patients may cause harm such as increased body dissatisfaction, unhealthy weight control behaviours and negatively impacting patient-provider relationships [51].

Therefore, to improve the quality of future studies, enhance the accuracy when testing the associations between IWB and health outcomes, and ensure such topics are approached in a safe and sensitive way, a tool specifically designed for children and youth is needed. The results of this review highlights three key components of weight internalisation that should be considered; 1) the lived experience and awareness of negative stereotypes 2) the acceptance and agreement of said stereotypes and 3) applying said beliefs to oneself and proceeding to mistreat or devalue oneself as a result [3]. Additionally, it will allow for a higher quality synthesis of findings for future systematic reviews and meta-analyses [21].

Psychological outcomes

This review identified that IWB held significant positive associations with multiple adverse psychological outcomes. It also identified the need for additional research across the weight spectrum, with studies finding that moderate to strong correlations between IWB and negative psychological health outcomes (e.g. eating disorder symptomology) persisted even after controlling for BMI [12, 52] and held true across weight categories [33]. Pakpour et al., found IWB to be more predictive of psychosocial disturbance and eating disorder symptomology than physical weight [46]. This may be due to the fact physical weight does not take into consideration the complex psychosocial complexities of self-worth/image. Hence, IWB may be more appropriate to screen for in a clinical settings to identify children and youth at risk of developing adverse outcomes rather than BMI alone [46]. However, additional research is required to support these associations using specialised tools.

Of note, preliminary clinical trial studies in adult populations found psychotherapy approaches to be effective in reducing IWB [53, 54], whilst others found that the theory of planned behaviours, health action process approach, and cognitive behavioural therapy were potential techniques to reduce self-stigma [55]. However, none have been trialled in a younger population.

Our review found IWB to have a role as a mediator in the relationship between weight stigma experiences, disordered eating behaviours [24] and psychosocial problems [33] which held true across weight categories. This agrees with findings from studies conducted with adult populations [50]. Our findings suggest that it may be how children and youth intrinsically manage their experiences, beliefs, and physical weight that leads to the development of harmful outcomes through IWB rather than physical weight alone.

Physical outcomes

Despite the above findings, it is important to recognise that many studies found strong positive associations between increased BMI and IWB levels [21, 24, 32–36,

38, 46]. Many only found this relationship to be true when comparing between groups with a BMI ≥ 25 kg/m² to groups with a BMI < 25 kg/m² [32, 34, 35, 44, 46] but not within populations with a BMI ≥ 25 kg/m² [36, 45]. This is in agreement with previous adult population studies that found no association between current BMI and IWB within populations with a BMI ≥ 25 kg/m² [10]. This may indicate that despite strong associations between higher BMI and IWB scores, the negative influence of IWB reaches a threshold beyond a particular weight. Hence, the extent of higher weight does not explain additional variance in IWB suggesting that individuals who have a higher BMI may internalise weight bias independent of physical weight [12, 36].

Currently, there is a lack of evidence to support that a decrease in BMI will alleviate IWB and its associated negative outcomes. Similarly, while the relationship between negative outcomes and IWB is clear, there is a lack of evidence to support that improving IWB will improve outcomes. However, this review has identified a need for further investigation into children and youth-specific treatment methods. Intervention and cause-effect experimental studies should be utilised with additional research into how IWB can be incorporated into health care practices. Future research should also aid health care professionals to recognise signs of IWB to help identify at-risk children and youth, allowing for early intervention and for the potential prevention of adverse health outcomes that have the tendency to endure into adulthood.

Strengths and limitations

Strengths of this study include following an explicit methodology with strict eligibility criteria and detailed search strategy to ensure specificity of the topic and allow for transparency and reproduction. It is also the first to investigate the association between IWB and health outcomes in a young population, however, it is not without limitations. Only one reviewer completed the quality assessment indicating potential for reporter bias. Additionally, for the purpose of this paper and due to the heterogeneity of the data, all eating disorder symptomologies were categorised under the same category. This does not take into consideration the complex multifactorial nature of different eating disorder symptomologies or their transdiagnostic properties [56]. Finally, due to the lack of robust data, a meta-analysis was unable to be completed.

Future direction

This review identified several gaps in current research. This research has contributed to understanding the groundwork for further research into IWB in children and youth to aid in the early identification and prevention of adverse health outcomes prior to them compounding into adulthood, decreasing the risk of lifelong adversities [57].

Firstly, most research was conducted with youth populations, with minimal studies undertaken with children. This could be due to a dearth of validated tools for children, ethics involved in conducting research with children and that they may not be advanced enough cognitively to understand the concepts of internalised weight bias. Children require separate investigation from youth due to fundamental biological, social and hormonal changes occurring during the development into adolescence, which limit the generalisability of our findings [29].

Secondly, a lack of racial and ethnic diversity was evident, with 12 articles not reporting any data on the ethnicity of participants. All studies within western populations were conducted on predominantly white populations. Studies that were conducted on different ethnicities (Middle Eastern and Asian) were unable to be generalised across countries due to cultural differences with the research not adequately capturing the experiences of varying populations. As child development models are not universal but are socially and culturally specific [58], generalisability between nations is difficult. Future research should include more diverse samples in assessments of IWB as it relates to health. Sex-related pathways additionally require further consideration given conflicting evidence on the role that gender contributes to IWB [35]. Finally, in comparison to psychological outcomes, physical and social outcomes were studied considerably less. Greater research should be conducted on these areas using objective measures [59].

Lastly, the cross-sectional nature of these studies do not allow for the formation of a cause-and-effect relationship. More prospective studies are required to investigate the bidirectional relationship between IWB and biopsychosocial outcomes and determine causality. Current research postulates that adverse outcomes may not only be consequences of IWB but also predisposing factors [8, 32], demonstrating a reciprocal predictive relationship, with bidirectional compounding occurring.

Changes to IWB move beyond the individual alone and span to the persons direct environment, family friends and society more broadly. It is however, recognised that changes at a societal level require time, and research such as this review can contribute to our understanding of the relationships in this area. Hence, this research contributes greater understanding to help provide the groundwork for further research into IWB in children and youth to aid in the early identification and prevention of adverse health outcomes prior to them compounding into adulthood, decreasing the risk of lifelong adversities [57].

Conclusion

This study aimed to identify associations between IWB and adverse health outcomes to inform health professionals of the importance of screening for IWB alongside BMI to identify a youth and children at risk of developing negative health outcomes as a result of IWB. IWB may be more clinically relevant in assessing risk in a population under 25 than physical weight alone due to its psychosocial aspects and ability to transcend beyond BMI. Research would benefit from better assessment tools designed for children and youth populations that accurately measure IWB. Psychotherapies may be an appropriate strategy in addressing IWB given success in adult trials, but research is required for younger populations. This review has contributed to the growing body of literature to help guide future interventions around identifying and comprehending IWB in children and youth.

Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s40337-023-00959-w.

Additional file 1: PRISMA reporting checklist.

Author contributions

TF was responsible for the conceptualisation, methodology, primary data collection, primary analysis and interpretation of results and draft manuscript preparation. ME aided in developing the search strategy. ME and YP completed full-text screening and contributed to the study design, data analysis, and interpretation of results. All authors read and approved the final version of the manuscript.

Funding

The Multiple Sclerosis Australia Fellowship supports Y.P. The other authors received no financial support for the research, authorship, or publication of this article.

Availability of data and materials

Datasets can be accessed via direct contact to corresponding author.

Declarations

Ethics approval and consent to participate Not applicable.

Competing interests

The authors have no conflicts or competing interests to declare.

Author details

¹School of Medical, Indigenous and Health Sciences, University of Wollongong, Wollongong, NSW, Australia.

Received: 4 July 2023 Accepted: 20 December 2023 Published online: 15 March 2024

References

 Puhl RM, Moss-Racusin CA, Schwartz MB. Internalization of weight bias: implications for binge eating and emotional well-being. Obesity. 2007;15(1):19–23.

- Pearl RL. Weight bias and stigma: public health implications and structural solutions. Soc Issues Policy Rev. 2018;12(1):146–82.
- Corrigan PW, Watson AC, Barr L. The self-stigma of mental illness: implications for self-esteem and self-efficacy. J Soc Clin Psychol. 2006;25(8):875–84.
- Puhl RM, Himmelstein MS, Quinn DM. Internalizing weight stigma: prevalence and sociodemographic considerations in US adults—internalizing weight stigma. Obesity. 2018;26(1):167–75.
- Latner JD, O'Brien KS, Durso LE, Brinkman LA, MacDonald T. Weighing obesity stigma: the relative strength of different forms of bias. Int J Obes. 2008;32(7):1145–52.
- Latner JD, Barile JP, Durso LE, O'Brien KS. Weight and health-related quality of life: the moderating role of weight discrimination and internalized weight bias. Eat Behav Int J. 2014;15(4):586–90.
- Pearl RL, White MA, Grilo CM. Weight bias internalization, depression, and self-reported health among overweight binge eating disorder patients. Obesity. 2014;22(5):E142–8.
- Ratcliffe D, Ellison N. Obesity and internalized weight stigma: a formulation model for an emerging psychological problem. Behav Cogn Psychother. 2015;43(2):239–52.
- Tyika TL, Annunziato RA, Burgard D, Daníelsdóttir S, Shuman E, Davis C, et al. The weight-inclusive versus weight-normative approach to health: evaluating the evidence for prioritizing well-being over weight loss. J Obes. 2014;2014(2014):983495–518.
- Pearl RL, Puhl RM. Weight bias internalization and health: a systematic review: weight bias internalization and health. Obes Rev. 2018;19(8):1141–63.
- 11. Puhl R, Suh Y. Stigma and eating and weight disorders. Curr Psychiatry Rep. 2015;17(3):552.
- 12. Durso LE, Latner JD. Understanding self-directed stigma: development of the weight bias internalization scale. Obesity. 2008;16(S2):S80–6.
- Lowes J, Tiggemann M. Body dissatisfaction, dieting awareness and the impact of parental influence in young children. Br J Health Psychol. 2003;8(2):135–47.
- 14. Volpe U, Tortorella A, Manchia M, Monteleone AM, Albert U, Monteleone P. Eating disorders: What age at onset? Psychiatry Res. 2016;238:225–7.
- 15. Harter S, Bukowski WM. The construction of the self developmental and sociocultural foundations. 2nd ed. New York: Guilford Press; 2012.
- Diedrichs PC, Puhl R. Weight bias: prejudice and discrimination toward overweight and obese people. In: Sibley CG, Barlow FK, editors. The Cambridge handbook of the psychology of prejudice. New York: Cambridge University Press; 2017. p. 392–412.
- Neumark-Sztainer DP, Wall MP, Larson NI, Eisenberg ME, Loth KM. Dieting and disordered eating behaviors from adolescence to young adulthood: findings from a 10-year longitudinal study. J Am Diet Assoc. 2011;111(7):1004–11.
- Nip ASY, Reboussin BA, Dabelea D, Bellatorre A, Mayer-Davis EJ, Kahkoska AR, et al. Disordered eating behaviors in youth and young adults with type 1 or type 2 diabetes receiving insulin therapy: the SEARCH for diabetes in youth study. Diabetes Care. 2019;42(5):859–66.
- Kuntsi J. Rutter's child and adolescent psychiatry. In: Thapar A, Pine DS, Leckman JF, Scott S, Snowling MJ, Taylor E, editors. 6th ed. Chichester: Wiley; 2015. pp. 1078, £135.00 (hb). ISBN: 978-1-118-38196-0 (Child and adolescent mental health. 2016;21(1):75-).
- Buttitta M, Iliescu C, Rousseau A, Guerrien A. Quality of life in overweight and obese children and adolescents: a literature review. Qual Life Res. 2014;23(4):1117–39.
- Ciupitu-Plath C, Wiegand S, Babitsch B. The weight bias internalization scale for youth: validation of a specific tool for assessing internalized weight bias among treatment-seeking German adolescents with overweight. J Pediatr Psychol. 2018;43(1):40–51.
- Schvey NA, White MA. The internalization of weight bias is associated with severe eating pathology among lean individuals. Eat Behav Int J. 2015;17:1–5.
- 23. Levy BR, Pilver CE. Residual stigma: psychological distress among the formerly overweight. Soc Sci Med. 2012;75(2):297–9.
- O'Brien KS, Latner JD, Puhl RM, Vartanian LR, Giles C, Griva K, et al. The relationship between weight stigma and eating behavior is explained by weight bias internalization and psychological distress. Appetite. 2016;102:70–6.

- 25. Hunger JM, Major B. Weight stigma mediates the association between bmi and self-reported health. Health Psychol. 2015;34(2):172–5.
- Kamolthip R, Fung XCC, Lin CY, Latner JD, O'Brien KS. Relationships among physical activity, health-related quality of life, and weight stigma in children in Hong Kong. Am J Health Behav. 2021;45(5):828–42.
- Salas XR, Forhan M, Caulfield T, Sharma AM, Raine KD. Addressing internalized weight bias and changing damaged social identities for people living with obesity. Front Psychol. 2019;10:1409.
- 28. Child CotD. Early Childhood Mental Health; 2013.
- Shiner R, Caspi A. Personality differences in childhood and adolescence: measurement, development, and consequences. J Child Psychol Psychiatry. 2003;44(1):2–32.
- Page MJM, Joanne E, Bossuyt PM, Boutron I, Hoffmann TC, Mulrow CD, Shamseer L, Tetzlaff JM, Akl EA, Brennan SE, Chou R, Glanville J, Grimshaw JM, Hróbjartsson A, Lalu MM, Li T, Loder EW, Mayo-Wilson E, McDonald S, McGuinness LA, Stewart LA, Thomas J, Tricco AC, Welch VA, Whiting P, Moher D. The PRISMA 2020 statement: an updated guideline for reporting systematic reviews. BMJ. 2021;372:n71.
- 31. Association AD. ADA quality criteria checklist: primary research; 2012.
- Gmeiner MS, Warschburger P. Intrapersonal predictors of weight bias internalization among elementary school children: a prospective analysis. BMC Pediatr. 2020;20(1):408.
- Zuba A, Warschburger P. The role of weight teasing and weight bias internalization in psychological functioning: a prospective study among school-aged children. Eur Child Adolesc Psychiatry. 2017;26(10):1245–55.
- 34. Gmeiner MS, Warschburger P. Simply too much: the extent to which weight bias internalization results in a higher risk of eating disorders and psychosocial problems. Eat Weight Disord. 2021;27(1):317–24.
- Puhl RM, Himmelstein MS. Weight bias internalization among adolescents seeking weight loss: implications for eating behaviors and parental communication. Front Psychol. 2018;9:2271.
- 36. Roberto CA, Sysko R, Bush J, Pearl R, Puhl RM, Schvey NA, et al. Clinical correlates of the weight bias internalization scale in a sample of obese adolescents seeking bariatric surgery. Obesity. 2012;20(3):533–9.
- Romano KA, Heron KE, Henson JM. Examining associations among weight stigma, weight bias internalization, body dissatisfaction, and eating disorder symptoms: Does weight status matter? Body Image. 2021;37:38–49.
- Himmelstein MS, Puhl RM. Weight-based victimization from friends and family: implications for how adolescents cope with weight stigma. Pediatr Obes. 2019;14(1):1–11.
- Fan C-W, Liu C-H, Huang H-H, Lin C-Y, Pakpour AH. Weight stigma model on quality of life among children in Hong Kong: a cross-sectional modeling study. Front Psychol. 2021;12:629786.
- Ra J, Kim, SO factors associated with the weight bias internalization of the girls in early adolescence. J Korean Acad Community Health Nurs; 2021.
- Ahorsu DK, Lin CY, Imani V, Griffiths MD, Su JA, Latner JD, et al. A prospective study on the link between weight-related self-stigma and binge eating: Role of food addiction and psychological distress. Int J Eat Disord. 2020;53(3):442–50.
- 42. Lin C-Y, Imani V, Broström A, Huus K, Björk M, Hodges EA, et al. Psychological distress and quality of life in Iranian adolescents with overweight/ obesity: mediating roles of weight bias internalization and insomnia. Eat Weight Disord. 2019;25(6):1583–92.
- O'Hara L, Tahboub-Schulte S, Thomas J. Weight-related teasing and internalized weight stigma predict abnormal eating attitudes and behaviours in emirati female university students. Appetite. 2016;102:44–50.
- 44. Wong PC, Hsieh YP, Ng HH, Kong SF, Chan KL, Au TY, Lin CY, Fung XC. Investigating the self-stigma and quality of life for overweight/ obese children in Hong Kong: a preliminary study. Child Indic Res. 2018;12(3):1065–82.
- Maïano C, Aimé A, Lepage G, Morin AJS. Psychometric properties of the weight self-stigma questionnaire (WSSQ) among a sample of overweight/obese French-speaking adolescents. Eat Weight Disord. 2017;24(3):575–83.
- 46. Pakpour AH, Tsai M-C, Lin Y-C, Strong C, Latner JD, Fung XCC, et al. Psychometric properties and measurement invariance of the weight self-stigma questionnaire and weight bias internalization scale in children and adolescents. Int J Clin Health Psychol. 2019;19(2):150–9.

- Saffari M, Chen J-S, Wu H-C, Fung XCC, Chang C-C, Chang Y-L, et al. Effects of weight-related self-stigma and smartphone addiction on female university students' physical activity levels. Int J Environ Res Public Health. 2022;19(5):2631.
- Lin C-Y, Strong C, Latner JD, Lin Y-C, Tsai M-C, Cheung P. Mediated effects of eating disturbances in the association of perceived weight stigma and emotional distress. Eat Weight Disord. 2020;25(2):509–18.
- Purton T, Mond J, Cicero D, Wagner A, Stefano E, Rand-Giovannetti D, et al. Body dissatisfaction, internalized weight bias and quality of life in young men and women. Qual Life Res. 2019;28(7):1825–33.
- 50. Meadows A, Higgs S. A bifactor analysis of the weight bias internalization scale: What are we really measuring? Body Image. 2020;33:137–51.
- Loth KA, Lebow J, Uy MJA, Ngaw SM, Neumark-Sztainer D, Berge JM. First, do no harm: understanding primary care providers' perception of risks associated with discussing weight with pediatric patients. Global Pediatric Health. 2021;8:2333794X211040979.
- Puhl RM, Lessard LM, Himmelstein MS, Foster GD. The roles of experienced and internalized weight stigma in healthcare experiences: perspectives of adults engaged in weight management across six countries. PLoS ONE. 2021;16(6):e0251566.
- Palmeira L, Pinto-Gouveia J, Cunha M. Exploring the efficacy of an acceptance, mindfulness & compassionate-based group intervention for women struggling with their weight (Kg-Free): a randomized controlled trial. Appetite. 2017;112:107–16.
- Pearl RL, Hopkins CH, Berkowitz RI, Wadden TA. Group cognitive-behavioral treatment for internalized weight stigma: a pilot study. Eat Weight Disord. 2016;23(3):357–62.
- Cheng OY, Yam CLY, Cheung NS, Lee PLP, Ngai MC, Lin C-Y. Extended theory of planned behavior on eating and physical activity. Am J Health Behav. 2019;43(3):569–81.
- Fairburn CG, Cooper Z, Shafran R. Cognitive behaviour therapy for eating disorders: a "transdiagnostic" theory and treatment. Behav Res Ther. 2003;41(5):509–28.
- 57. Eisenberg ME, Neumark-Sztainer D, Haines J, Wall M. Weight-teasing and emotional well-being in adolescents: longitudinal findings from project EAT. J Adolesc Health. 2006;38(6):675–83.
- Punch S. Research with children: The same or different from research with adults? Childhood. 2002;9(3):321–41.
- Cramer H. Public health: a multi-faceted issue. J Public Health. 2013;21(2):121–2.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.